Matrices & Linear Algebra I B Term, Sections C01-C04 W. J. Martin February 4, 2002

Week 5: Reading and Exercises

Reading

Please read Sections 6.3, 6.4 and 6.5 this week. You will have noticed that concepts are now taking precedence over procedures. So I again encourage you to read carefully, pause to think now and then, and always havea pad of paper and pencil nearby to work out the details which may be omitted in the text.

Practice Exercises

Note: Do not hand in.

Here are some practice problems to work on at home. It is very important that you get started on these early.

If you have any questions about them, please discuss them in conference or come see me.

• page 261-2, # 3-11 (odd), 15, T.1, T.2, T.4, T.5.

• page 272-4, # 1-21 (odd), 29, 35, T.3, T.4, T.10, T.12.

• page 282-3, # 1-21 (odd).